Author’s response to reviews

Title: Macular hole surgery recovery with and without face-down posturing: a meta-analysis of randomized controlled trials

Authors:

Ting Ye (yetingpeace@163.com)
Ji-guo Yu (yjgyanke@163.com)
Lin Liao (voicemeeaa@163.com)
Lan Liu (258602524@qq.com)
Ting Xia (694832403@qq.com)
Lei-lei Yang (565422607@qq.com)

Version: 2 Date: 02 Dec 2019

Author’s response to reviews:

Dear Dr. Jashan Abraham,

We thank you again for your kindly and carefully review for my article. I have provided the clean version as requested. For the figures, I hope to keep the colors to help readers better understand the meaning of the figures.

We hope that the revised manuscript is now suitable for publication in your journal, and we look forward to hearing from you.

Yours sincerely,

Lei-lei Yang

Department of Ophthalmology, Wuhan Fourth Hospital, Puai Hospital, Tongji Medical College, Huazhong University of Science and Technology, Wuhan 430033, Hubei Province, China.
Tel: + 86 027 68831509, Fax: + 86 027 68831509, 565422607@qq.com